

#### Centre for Corporate Governance Research

Research Report No. 1/2025

January 2025

## **An Overview of Norwegian Firms**

**Janis Berzins and Bogdan Stacescu** 

© Janis Berzins and Bogdan Stacescu. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission, provided that full credit, including © notice, is given to the source. This paper can be downloaded without charge from the CCGR website http://www.bi.edu/ccgr



#### Table of Contents

1.	In	ntroduction	3		
2.	Ke	ey statistics	5		
3.	Da	Pataset	18		
4.	Ge	eneral overview of Norwegian limited liability firms	19		
4.	1.	Limited liability firms by industry	22		
5.	Lis	isted and nonlisted firms	27		
6.	Ne	lew firms	37		
7.	Fa	amily and nonfamily firms	44		
7.1. The frequency of family firms. The share of family firms in employment, revenues and					
as	sse	ets	45		
7.	2.	The size distribution of family and nonfamily Norwegian firms	47		
7.	3.	The industry distribution of family and nonfamily Norwegian firms	49		
8.	Fii	inancial indicators by firm size, family and nonfamily firms	52		
9.	Fii	inancial indicators over time, family and nonfamily firms	75		
9.	2.	Financial indicators over time, large family and nonfamily firms	90		
10.		Financial indicators for family and nonfamily firms by industry, 2000-2023	101		
11.		The governance of limited liability Norwegian firms	121		
11	1.2.	2. Ownership	121		
11	1.3.	3. The board of directors	123		
11	1.4.	I. CEOs	126		
12.		Governance: Family and nonfamily firms	130		
12	2.2.	2. Ownership	130		
12	2.3.	3. The board of directors	131		
12	2.4.	l. CEOs	136		
13.		Governance and firm size	139		
13	3.2.	2. Ownership	139		
13	3.3.	3. The board of directors	141		
13	3.4.	l. CEOs	148		
14.		Governance: Large family and nonfamily firms	154		
14	1.2.	2. Ownership	154		
14	1.3.	3. The board of directors	155		
14	1.4.	l. CEOs	162		
15.		Family firms with/without minority investors and family CEOs	166		
16.		Foreign ownership	196		
17.		Corporate and personal finance	201		

17.2.	Wealth taxation over time	201		
17.3.	Wealth taxation and firm characteristics	207		
18. L	ocation, firms and owners	227		
18.1.0	Centrality and family firms	227		
18.2.	Fish farming: an industry for less central regions			
18.3.	Centrality and foreign ownership	250		
References				
Appendix 2				
Variab	le definition list			
Financial indicators for family and nonfamily firms by industry, 2023				

## 1. Introduction

Norway is one of the wealthiest countries in the world, with the 3<sup>rd</sup>-largest GDP per capita in the OECD and the second-highest level of labor productivity<sup>1</sup>. At the same time, the Norwegian economy is facing important challenges, such as an ageing population and the green transition. It is therefore important to take stock and examine the current state of Norwegian firms. In this report we present an overview of their corporate governance characteristics and financial performance.

Our study starts from the universe of limited liability firms registered in Norway. It covers the period between 2000 and 2023. We group firms in business groups and present consolidated numbers for the group. That leaves us with a population that increases from around 100,000 limited liability firms (groups) in 2000 to 250,000 in 2023.

The vast majority (more than 99%) of the firms in our study are private firms. Our dataset therefore allows us to provide a detailed picture of nonlisted firms – firms that are largely missing from the finance and governance research that usually uses data reported by listed firms. We show that private firms represent a large proportion of economic activity, and that their governance characteristics are different.

The population of limited liability firms includes many very small firms, and a few very large ones. This heterogeneity is important, and we also present separate statistics for medium and large firms.

Firms in different industries can be quite different in terms of typical size, capital intensity, profitability, growth and even governance. We therefore also present industry-level statistics at several points in our study.

Norway is a country where the population is spread out over a relatively large area, and the balance between larger cities and small places is quite important. Our study examines the characteristics and overall importance of firms headquartered in more or less central regions, and presents the differences in terms of size, performance, ownership, and governance. We also look at the importance of foreign ownership across firm sizes, industries, and regions.

The vast majority of nonlisted firms are controlled by individuals and families. Family firms can be quite different from other firms in terms of their governance, financing, growth trajectories, and even financial performance. We present a detailed set of statistics on family firms throughout our report<sup>2</sup>.

Family firms tend to be smaller on average than nonfamily firms – even though large family firms do exist. It is therefore important to try to distinguish between family firm characteristics that are driven by their ownership and those that are driven by their typically smaller size. Family firms are also more important in some industries than others, and represent a larger proportion of revenues, assets and employment in less central areas. To address these important sources of heterogeneity, we present separate statistics for family and nonfamily firms in different size brackets, industries, and geographies.

<sup>&</sup>lt;sup>1</sup> OECD (2024), *OECD Compendium of Productivity Indicators 2024*, OECD Publishing, Paris, <u>https://doi.org/10.1787/b96cd88a-en</u>.

 $<sup>^{2}</sup>$  We define family firms as firms where at least 50% of the equity is ultimately controlled by a family or a group of individuals who are related by blood or marriage.

Our report starts with a summary of key statistics in Section 2. We present details about the structure and sources of our dataset in Section 3. In section 4 we present general characteristics of Norwegian limited liability firms and differences across industries. Section 5 presents statistics on listed Norwegian firms, and compares them to the much larger set of private firms. In section 6, we look at the cohorts of new firms started during our sample period, and examine their performance during the first years of their existence.

Section 7 begins the exploration of family firms. It presents the proportion of family firms in the total number, revenues, assets and employment of limited liability firms. It describes the size distribution of family and nonfamily firms, and the importance of family firms across industries. Section 8 presents an overview of financial indicators for family and nonfamily firms of different sizes, while sections 9 and 10 present financial indicators over time and across industries respectively.

The corporate governance of various firms can be an important factor in their performance. Corporate governance structures can also be very different between family and nonfamily firms. Section 11 presents a basic overview of the ownership, board of directors and CEO characteristics for Norwegian limited liability firms. Section 12 compares the governance characteristics of family and nonfamily firms, section 13 links governance to firm size, and section 14 focuses on large family and nonfamily firms.

Some family firms also have minority shareholders, even if some of them do not. Most family firms have family CEOs – but some of them choose to have nonfamily CEOs. Section 15 presents an overview of family firms with those different characteristics.

Foreign ownership is important on the Oslo stock market – and on Nordic markets in general, given that they are small open economies. Section 16 provides an overview of foreign ownership including private firms in various industries.

In firms with concentrated ownership, and in particular in family firms, there is a close connection between the finances of the firms and the owners' personal resources. Section 17 presents a brief overview of that connection.

Geography can be an important factor for the size, financial performance, investor base and governance of firms. Section 18 presents an overview of firms in different geographical regions based on their centrality. It compares firms headquartered in large cities to firms in smaller and more remote communities. It showcases the fish farming industry as an industry that is well represented in less central regions, and also presents the importance of foreign ownership across regions.

## 2. Key statistics

The **number** of active Norwegian **limited liability firms**<sup>3</sup> has increased significantly in recent years, following a decrease in the amount of required equity and the relaxation of auditing requirements for small limited liability firms in 2011<sup>4</sup>.



Since some small firms that were/would have been unlimited liability firms became limited liability firms, the average firm **size** has decreased slightly in recent years. The graph below presents the mean and median revenues (sales) of Norwegian limited liability firms in million NOK. The amounts are inflation adjusted and expressed in 2023 kroner.



Given the increase in small limited liability firms, the proportion of **listed firms**, always quite low, has decreased slightly. Moreover, the share of listed firms in the total revenues of Norwegian limited liability firms has stabilized at a lower level since the oil price shock of 2015.

<sup>&</sup>lt;sup>3</sup> We start from the universe of active limited liability firms in Norway. One business group counts as one observation, and we use consolidated data for business groups.

<sup>&</sup>lt;sup>4</sup> The minimum amount of equity needed to start a limited liability firm (AS) was reduced from 100,000 to 30,000 kroner. Also, from 2012 the firm can choose to not have an auditor if sales are below NOK 6 mill., assets are below NOK 23 mill., and the average number of employees during the year is below 10.





**Family firms**<sup>5</sup> represent a majority of Norwegian limited liability firms. The increase in the number of small limited liability firms is also associated with an increase in the number and even the proportion of limited liability family firms.



Left scale: number of firms. Right scale: the proportion of family firms in limited liability Norwegian firms in a given year.

<sup>&</sup>lt;sup>5</sup> In this study family firms are defined as firms where an individual or a group of individuals related by blood and marriage ultimately control at least 50% of the equity. Additional sample and variable definitions can be found in the Appendix.

Even though they are on average smaller than nonfamily firms, family firms also represent a significant proportion of the overall employment, revenues, and assets in limited liability firms, and the proportions has been fairly stable over time:



Family firms also play a major role across industries (with lower proportions in capital intensive industries such as mining and oil and utilities).



Family firms are dominant among small and medium enterprises, but they are also present among the largest firms in the economy. In the graph below, we group firms into size deciles based on their revenues in 2023, with 1 being the smallest and 10 being the largest firms. We calculate the proportion represented by family firms in each size decile.



Family firms have **higher profitability and lower growth** compared to nonfamily firms, and the difference seems to be stable across the business cycle.



(The operating return on assets is calculated as the ratio of after-tax operating earnings to total assets, winsorized at 2.5% and 97.5%.)



(The real growth of sales in year t is calculated as  $(Sales_t-Sales_{t-1})/Sales_{t-1}$ , where sales (revenues) are inflation adjusted and expressed in 2023 kroner. The ratio is winsorized at 97.5%.)

Norway has a relatively small population that is spread out across regions with varying densities and connectivity to the rest of the country. Family firms represent a larger share of firms in **less** central regions, as shown in the graph below<sup>6</sup>.



Left scale: the number of firms. Right scale: the proportion of family firms.

A significantly larger share of the employment, revenues, and assets of family firms is located in less central areas. The graphs below show the proportion of employees, revenues, and assets in firms headquartered in municipalities with a given centrality class, for family and nonfamily firms respectively.



<sup>&</sup>lt;sup>6</sup> We use the centrality index from Statistics Norway

<sup>(&</sup>lt;u>https://www.ssb.no/befolkning/folketall/artikler/sentralitetsindeksen</u>) to classify firms and business groups based on the location of their headquarters. 1 is the Oslo region (the most central), 2 includes other large cities (Bergen, Trondheim, Stavanger, more distant municipalities in the Oslo region), 6 represents the least central municipalities (e.g. Karasjok, Dovre, Moskenes or Ulvik).



There are industries that are significantly better represented in less central regions, such as fish farming. Around half (between 40% and 60%, depending on centrality index) of fish farming firms are family firms.



The typical **size** of nonfamily firms decreases as we go from more to less central locations. That is not the case for family firms – which means that the size difference between family and nonfamily firms is much smaller in less central regions.



In a majority of family firms, the family controls 100% of the shares. The typical **number of owners** is significantly larger in nonfamily than in family firms in all size brackets (1 = smallest decile of firms by revenues in 2023, 10=the largest decile).



Norway pioneered the introduction of **gender balance** requirements in large (ASA) firms. Looking at all limited liability firms – not just the largest ones – the proportion of women on board has been larger in family firms throughout our sample period, but it is still far from balanced. The proportion of women on board is also higher in family firms across the firm size distribution – but it is still slightly lower in the largest firms.





The proportion of women CEOs has been increasing over time, but it is still quite low. It is higher in family firms.





The proportion of women entrepreneurs<sup>7</sup> has also been slowly increasing over time, in line with similar evolutions in other Scandinavian countries.

In Norway, as in most other countries, the population has been growing **older** in recent decades. That is also reflected in an increasing typical age of firm insiders (directors and CEOs).



Controlling families are usually **closely involved** in the family firm. Even among firms in the largest size decile, the chair of the board and the CEO are family members in most cases.

<sup>&</sup>lt;sup>7</sup> In this analysis we define entrepreneurs as the CEOs of newly established limited liability firms that also control at least 50% of the firms' equity.



The equity financing of family firms comes mostly from the family, and controlling families are usually not well diversified regardless of the size of the family firm. To illustrate the resulting close connection between firm and personal finances, we form the following groups of family businesses based on the net wealth of the controlling family:

- Group 0: negative net wealth;
- Group 1: net wealth between 0 and 500,000 kroner (adjusted for inflation, in 2023 kroner);
- Group 2: net wealth between 500,000 and 1,000,000 kroner;
- Group 3: net wealth between 1,000,000 and 5,000,000 kroner;
- Group 4: net wealth between 5,000,000 and 10,000,000 kroner;
- Group 5: net wealth between 10,000,000 and 50,000,000 kroner;
- Group 6: net wealth between 50,000,000 and 100,000,000 kroner;
- Group 7: net wealth above 100,000,000 kroner.

The size distribution of the firms follows closely the distribution of the controlling family's wealth, indicating the lack of personal diversification and the limited role of personal and corporate leverage.



The net wealth is also the tax base for the wealth tax. We calculate the ratio between the controlling family's annual wealth tax payment and its personal holdings of liquid assets (bank savings and listed securities). We can see that wealthier business owners also pay more in the wealth tax relative to their personal liquid assets. The ratio has increased in the second half of our time period.



**Foreign ownership** represents an important type of funding and governance. It is also well represented in Norway given its status as a small open economy. We define foreign-controlled firms as firms where foreign individuals or companies ultimately control more than 50% of the equity. The proportion of firms with foreign majority ownership is highest in mining and oil and IT. It is lower in fishing and farming.





In terms of revenues, foreign-controlled firms are important in services, mining and oil, and IT.

Firms controlled by non-Norwegian individuals or corporations represent a relatively higher proportion in more central regions.



Foreign-controlled firms tend to be larger and to represent a larger proportion of revenues in more central regions.



## 3. Dataset

Our starting dataset consists of all limited liability firms registered in Norway between 2000 and 2023. Accounting, ownership, and board data are delivered by Experian (www.experian.no). Data on family relationships are from Folkeregisteret (www.skatteetaten.no/en/person/national-registry), which stores the census data. These two data sets are organized as one integrated database by the CCGR, which is the Centre for Corporate Governance Research (www.bi.edu/ccgr). Statistics Norway (www.ssb.no/en ) and Skattedirektoratet (www.skatteetaten.no/en/person) have delivered the data on shareholder wealth and income from the tax returns as well as data on individual salary and dividends income.

Every Norwegian firm with limited liability is legally obliged to publish full accounting statements every year. The firm must also report the identity of its CEO, directors, and owners, as well as each owner's equity holding in the firm. These data are submitted to a state agency (*Brønnøysundregistrene*; www.brreg.no/home). Experian acquires the data.

We compute several main measures from the basic data to organize the data along two major dimensions: links between the firms and links between the owners.

We compute the ultimate ownership structure by looking up the shareholders of corporate owners, thus accounting for indirect ownership in every firm. The ultimate owner can only be an individual (local taxpayer or foreigner), state, financial institution, foreign firms, or a firm with unknown owners (domestic of foreign).

Building indirect ownership chains also enables us to control for the fact that many firms in our sample belong to business groups (*konsern*). That allows us to avoid the double counting that would happen if we considered group member firms as independent entities. A group in our data is constructed by examining majority ownership links between firms, where the majority link can also be indirect as long as it goes through another corporate entity. We find that the groups are up to ten levels deep.

We identify the firms in the group with most representative finance and governance reporting. These may be two different firms in the group. For financial variables, we use the consolidated values reported at the highest level in the group where consolidation occurs. In a few cases, that is not the level where we have the largest or most representative board, and in those cases we choose the alternative firm as representative for governance variables.

We group individuals (shareholders, directors and CEOs in a firm) into families using four vertical degrees of kinship, both up and down (e.g., a parent is up while a child is down) and two lateral degrees (e.g., cousins are second lateral degree). We define family firms (business groups) as firms (groups) where a family ultimately controls at least 50% of the equity.

# 4. General overview of Norwegian limited liability firms

The number of limited liability firms in Norway has increased in line with population growth and the growth of the economy, but also due to regulatory changes that reduced the minimum capital required to start a limited liability firm from 2012. The total went from around 100,000 in 2000 to almost 250,000.



The typical firm size, measured by revenues (sales), total assets, and employment has declined slightly as a result of the influx of relatively smaller firms. The size distribution is highly skewed (there are many small firms and a few very large firms)<sup>8</sup>.



<sup>&</sup>lt;sup>8</sup> The reporting rules for employment changed in 2015 (<u>https://www.skatteetaten.no/en/business-and-organisation/employer/the-a-melding/about-the-a-ordning/about-a-ordningen/</u>), hence the increase from previous years.



The *total* amount of revenues, total assets and employment in limited liability firms has increased gradually over time<sup>9</sup>.





<sup>&</sup>lt;sup>9</sup> Note that the amounts are inflation adjusted (expressed in million 2023 kroner), so this is growth in real terms.

The significant skewness of the distributions is also illustrated below. The distributions of the logs of revenues and assets is leptokurtic, with heavier tails (a higher proportion of very small and very large firms) than the normal distribution. The distribution of employment is also very highly skewed, with many firms with one employee and a few with large numbers of employees.







#### 4.1. Limited liability firms by industry

The Norwegian economy has a diversified structure, with some industries such as oil and gas, fishing and tourism playing an important role in its exports. We present an overview of firm characteristics by industry below. We group limited liability firms in 17 industries<sup>10</sup>.

The largest numbers of firms are in services, real estate, retail and wholesale trade, and construction.



<sup>&</sup>lt;sup>10</sup> "Public administration" includes services that are similar to the ones usually offered by the public sector, e.g. private kindergartens or clinics. "Multisector" includes firms and business groups that report industry codes from multiple industries.

However, those industries have many small firms. If we look at the distribution of total activity, measured by revenues (sales), total assets, and employment, the weight of multisector firms (firms and business groups that are active in multiple industries) increases. Some industries are more labor intensive and have a relatively larger share in employment (services, retail and wholesale), while others are more capital intensive and are more important in terms of total assets (real estate, mining and oil).





The firm-level averages confirm the skewed distributions (the large differences between means and medians), and the relatively large size of firms in mining and oil. Utility firms are obviously capital intensive.







The distribution of value added (total payroll plus total earnings) across industries confirms the importance of mining and oil, services and multisector firms.





#### 5. Listed and nonlisted firms

Recent research points to a decline in the number of listed firms, even on the successful and dynamic US markets (Doidge et al. 2017). The reasons behind this trend have been debated, with acquisitions and delistings aimed at gaining a more concentrated and stable investor base as partial explanations (Doidge et al. 2017, Boot et al. 2008). At the same time, there has been considerable growth in private markets, and many large firms tend to stay private for longer. Private firms have become larger, with more dispersed ownership and governance structures reminiscent of listed firms (Lowry 2023).



Listed firms represent a small proportion of firms in Norway. Given the increase in the overall number of limited liability firms in Norway and relatively stable number of listed firms, the proportion of listed firms has declined slightly in recent years:

Listed firms are much larger than nonlisted firms, and that is shown by their significant share in the total revenues of Norwegian limited liability firms. That share moved from around one third to around one fifth following the oil price shock of 2015.



Even though there are some smaller listed firms that are of comparable size to nonlisted firms, the typical listed firm is significantly larger:



Perhaps consistent with the idea of a blurring of the lines between listed and nonlisted firms, the overlap between the size distribution of the two groups has increased over time. The histograms below present the distribution of the log of sales (in million 2023 kroner) for listed firms (in red) and nonlisted firms (in blue), in 2023 and 2000 respectively. Comparing the two graphs we can see that the overlap between the two size distributions seems to have increased over time, perhaps consistent with the idea of a blurring of the lines between listed and private firms.





Listed firms also represent a large part of the total assets and employment in Norwegian firms:









Listed firms have represented between 20% and 35% of the value added by Norwegian limited liability firms, where value added is calculated as the sum of salaries and earnings. The value added is cyclical and it decreased in 2008, during the oil price shock of 2015, and during the Covid-19 crisis:





The ownership structure of private firms is much more concentrated than that of listed firms. Moreover, directors and managers are more likely to be large shareholders themselves.

We illustrate those differences in the graph below, which presents the share of the largest shareholder. Given that listed firms tend to be much larger than nonlisted firms, we also include large nonlisted firms as an additional category, where large firms are defined as firms (or business groups) with at least 100 employees and at least 100 million (2023) kroner in sales.

As expected, the share of the largest shareholder is much larger in private than in listed firms. Indeed, the median private firm is a single-owner firm: an individual or family controls 100% of the equity. Large private firms are very close to the typical nonlisted firm and quite far from the listed firms of similar size.



Unusually in comparison to other countries, listed Norwegian firms are characterized by having multiple large blockholders (Bøhren and Ødegaard 2000). The ownership share of the second-largest shareholder is fairly large in both listed and nonlisted firms.



Ownership by insiders (directors and CEOs) presents the advantage of stronger incentives, and the potential problem of entrenchment. We present statistics on insider ownership below.

We can see that the insider ownership share is much lower in listed than in private firms. Even though large nonlisted firms are closer in size to listed firms, they are much closer to the typical private firm in insider ownership.



If we look at just ownership by directors, there is once again a large difference between listed and nonlisted firms, with large nonlisted firms somewhere in the middle between listed and the typical private firm.



Norwegian boards tend to be small by international standards. The boards of large private firms tend to be smaller than the boards of listed firms, and both groups are much above the typical number of directors in a private firm. Board size is positively correlated with firm size.



Norway was the first country to impose a gender balance rule for the boards of corporations<sup>11</sup>, and that is reflected in the relatively high (and stable) proportion of women on the boards of listed firms after 2006. The proportion was higher in nonlisted firms prior to the quota, but it has lagged behind listed firms in more recent years. Interestingly, large private firms (with relatively large boards) have a lower proportion of women on board than the average private firm.



The typical director age has been increasing in recent years. Directors in larger firms, especially in listed firms, tend to be older.

<sup>&</sup>lt;sup>11</sup> An overview of the board reform and its impact can be found in Eckbo et al. (2022), Bøhren and Strøm (2010), Bøhren and Staubo (2016), Ahern and Dittmar (2012).



The higher ownership concentration in private firms is also associated with higher ownership by CEOs. Indeed, the CEO is the only shareholder in the median private firm in recent years. Even in large private firms, the CEO's average ownership share is twice the share in listed firms.



The proportion of female CEOs is quite low, and it has been increasing at a slow pace over the past two decades. It is significantly smaller in larger firms, particularly in listed firms, where the average proportion of female CEOs is half of the typical proportion in nonlisted firms.


Similarly to the age of directors, the average CEO age has also been on an increasing trend.



### 6. New firms

The establishment of new firms is often associated with innovation, risk taking, and economic growth (Audretsch 2018). Entrepreneurs are the ones that experiment with new ideas and can produce new businesses and new technologies (Kerr et al. 2014, Manso 2016). The recent EU Competitiveness <u>Compass</u> (January 2025) emphasizes the importance of entrepreneurship in achieving the bloc's goals and enhancing productivity growth.

At the same time, not all entrepreneurs are successful (Hamilton 2000), and some entrepreneurship, especially at the top of the wealth distribution, is less productive and has sometimes been considered a "luxury good" (Hvide and Møen 2010). Skills, risk taking, industry knowledge, peer effects and financial constraints are important success factors (Giannetti and Simonov 2009, Hall and Woodward 2010, Andersen and Nielsen 2012, Lerner and Malmendier 2013, Hvide and Panos 2014, Hvide and Oyer 2017, Hombert et al. 2020).

In this section, we present an overview of Norwegian new (entrepreneurial) firms during our sample period. We define new firms as newly registered firms and present them from their first year of activity<sup>12</sup>.

The number of new Norwegian limited liability firms has increased from around 5,000 per year in the early 2000s to around 15,000 in more recent years. The change occurs around the 2012 regulatory changes that made it easier to start a limited liability firm.



Firms started in the same year may share common characteristics and may be affected by similar macroeconomic shocks. There may also be significant differences between firms from cohorts started in different years. We present below some statistics that group new firms by cohorts and follow them over their first 5 years.

<sup>&</sup>lt;sup>12</sup> For an earlier analysis of entrepreneurship in Norway that also includes self-employment, see Berglann et al. (2011).

Many young firms fail within the first years. As shown in the next graph, almost half of new firms do not survive for at least 5 years. Survival rates are lower during downturns, such as the 2008 financial crisis.



The first years in a firm's life are also characterized by high growth, whether we measure that in terms of sales, total assets, or employment. Firm size and growth are similar if we just focus on the firms that survive for the first 5 years.







The profitability of new firms is quite variable and it can be strongly affected by economic shocks such as the 2008 financial crisis, the 2015 decrease in oil prices, or the Covid-19 crisis. The graphs below show the profitability measured as the operating return on assets (after-tax operating earnings divided by total assets). The profitability of surviving firms is significantly higher than that of the overall sample.





We can also look at the characteristics of entrepreneurs starting new firms. To do that, we restrict our attention to new firms where the CEO holds at least 50% of the shares – and those CEOs constitute our sample of entrepreneurs.



The number of entrepreneurs controlling limited liability companies has increased over time, and that applies to entrepreneurs of both genders.

The participation of certain groups such as women, young people or immigrants in entrepreneurship has been an important policy concern in recent years (OECD/European Commission (2023)). The proportion of women entrepreneurs has gradually increased over time, but it is still at 22% in 2023<sup>13</sup>.



<sup>13</sup> The proportion is similar in neighboring countries. Numbers on the proportion of women entrepreneurs in Denmark, for instance, can be found in the 2022 report from Danmarks Statistisk at <u>https://www.dst.dk/da/Statistik/nyheder-analyser-publ/Publikationer/VisPub?cid=40865</u>. Fjærli, Iancu, and Raknerud (2013) find similar percentages for Norway in the early 2000s using a slightly broader definition of entrepreneurship. The age distribution of entrepreneurs is quite broad and similar across genders, as shown by the histogram below (red and blue represent female and male entrepreneurs, respectively). Consistent with Berglann et al. (2011), we find that the average age of entrepreneurs is in the early 40s. The age distribution seems to be fairly stable over time.





The size distribution in the first year is quite similar for male and female entrepreneurs. The graphs below present the distributions of the logs of revenues and logs of total assets in the first year. The latter shows that the minimum capital requirements are binding for many firms<sup>14</sup>.



The profitability of new firms with male entrepreneurs is slightly higher than that of new firms with female entrepreneurs during their early years.



<sup>&</sup>lt;sup>14</sup> van Praag and Raknerud (2014) find that a higher amount of capital committed by entrepreneurs is associated with higher performance in startups.

## 7. Family and nonfamily firms

The majority of firms in an economy are controlled by individuals or families. Family ownership is more widespread among nonlisted firms, and that means that easily available information on them tends to be more limited. Most corporate finance and corporate governance research has focused on listed firms, which capture a large part of overall economic activity and provide regular reports to the market.

As we confirm below, however, family firms also represent a significant section of the economy. They stand for around 70% of limited liability firms in Norway. They also represent around 40% of employment in limited liability firms, around 25% of revenues, and 20% of total assets. All these proportions have been fairly stable over time. The Norwegian firms confirm the importance of family ownership that has been documented in international studies (Amit and Villalonga, 2014, Bertrand and Schoar 2006).

Family firms are different from the typical listed firm in several important ways (Bennedsen, Perez-Gonzalez, and Wolfenzon, 2010). Their ownership is more concentrated, the communication between owners and managers is easier, but the relationship with minority owers is an important concern. The families are usually active owners, and are often also involved on the board and in the firm's management (Bøhren et al. 2019). The high level of involvement can be associated with higher performance, especially when the founder is still present in the firm (Perez-Gonzalez 2006, Villalonga and Amit 2006a, Anderson and Reeb 2003). Conversely, the family is usually not well diversified and is exposed to the idiosyncratic risk of the firm (Fagereng et al. 2020). The limited financial resources of the family can make the firm vulnerable in case of large shocks to financial markets (Lins et al. 2013).

The controlling family often has a long-term view and tries to protect the socioemotional wealth incorporated in the family firm (Gómez-Mejía et al. 2007). That can provide the advantages of lower debt financing costs (Anderson et al. 2003) and of more stable relationships with employees (Sraer and Thesmar 2007). It can also raise challenges related to succession issues (Bennedsen et al. 2007, Mehrorta et al. 2013, Berzins and Zaldokas 2023).

There are many definitions of family firms used in various studies (O'Boyle et al. 2012). This variation is often caused by data availability (having data on just listed firms and choosing a low family ownership threshold as a result, having to rely on surnames to identify kinship), but can also be conceptual (defining firms based on ownership, on the involvement of owners in management, distinguishing between classical family firms and entrepreneurial firms).

In what follows, a family firm is defined as a firm where an individual or a group of individuals related by blood or marriage ultimately control at least 50% of the equity. Business groups count as one observation.

We compare family firms to nonfamily firms – the remaining limited liability firms in the economy, where no individual or group of related individuals has a majority<sup>15</sup>. Most nonfamily firms are owned by multiple Norwegian individuals and families, with no majority for any of them. A smaller share are ultimately controlled by the state, financial institutions, or foreign individuals,

<sup>&</sup>lt;sup>15</sup> A previous research report looking at Norwegian family firms over the period 2000-2015 can be found at <a href="https://www.bi.edu/research/research-centres/centre-for-corporate-governance-research/Publications/">https://www.bi.edu/research/research-centres/centre-for-corporate-governance-research/Publications/</a>.

companies or financial institutions. We show that family firms are different from nonfamily firms in terms of typical size, industry and geographical distribution, performance, and governance structures.

## 7.1. The frequency of family firms. The share of family firms in employment, revenues and assets

The proportion of family firms among Norwegian limited liability firms is around 70%. It has increased slightly in recent years due to the reduction in minimum capital and auditing requirements, which has resulted in more small firms being started as limited liability firms. The number of family firms has increased at a faster rate than that of nonfamily firms.



Left scale: the number of firms, right scale: the proportion of family firms among limited liability firms

Family firms represent a smaller, but still significant proportion among large firms, where large firms are defined as firms (or business groups) with at least 100 employees and at least 100 million (2023) kroner in sales.



Left scale: the number of firms, right scale: the proportion of family firms among limited liability firms

Family firms represent a majority of medium-sized firms, where medium-sized firms are defined as firms with at least 10 employees or 10 million (2023) kroner in sales that are not classified as large.



Left scale: the number of firms. Right scale: the proportion of family firms.

Family firms represent around one quarter of the revenues and total assets of limited liability firms, and around 40% of employment.



### 7.2. The size distribution of family and nonfamily Norwegian firms

Proportion family firms, employment

 20% 10% 0%

While family firms are smaller on average than nonfamily firms, they are still quite varied in terms of size. This section presents an overview.

The graphs below present the distribution of all Norwegian limited liability firms by their sales in 2023, and the same for log sales. The distribution is very skewed, with many small firms and a few very large firms.



We can also compare the distribution of revenues for family and nonfamily firms. The graph below presents the distribution of sales (in million kroner and logs of million kroner) in 2023 for family firms (blue) and nonfamily firms (red) respectively. The distribution of nonfamily firm revenues is also skewed, but it puts more weight on relatively larger firms.



## 7.3. The industry distribution of family and nonfamily Norwegian firms

Family firms represent the majority of firms in most industries, with relatively low shares only in capital-intensive industries such as mining and oil and utilities.



Left scale: the number of firms, right scale: the proportion of family firms among limited liability firms



Family firms represent a very large proportion of employment in farming, forestry, real estate, tourism, and construction, and a relatively small proportion in mining and oil and IT:

Family firms represent a majority of revenues in real estate, farming, and forestry, a large proportion in tourism, fishing, public administration (e.g. private kindergartens), construction, and a small proportion in IT, utilities, and mining and oil:



In terms of assets, family firms represent a large proportion in farming, forestry, fishing, tourism, real estate, services – and a low proportion once again in IT, utilities, and mining and oil:



Family firms are dominant among small firms in all industries, They still represent a majority of medium-sized companies in most industries, especially in retail and wholesale and construction. The proportion is relatively low in IT and utilities.



Left scale: the number of firms, right scale: the proportion of family firms among medium-sized limited liability firms

The proportion of family firms is significantly lower among large firms, defined as firms (or business groups) with at least 100 employees and at least 100 million (2023) kroner in sales. Family firms are less common among large mining and oil and IT companies, and better represented among real estate and multisector firms.



Left scale: the number of firms, right scale: the proportion of family firms among large limited liability firms

# 8. Financial indicators by firm size, family and nonfamily firms

The summary statistics in the previous section have shown that family firms dominate among small and medium firms, while still representing a significant proportion of the largest firms in the economy. Large family firms are still quite likely to be different from small family firms. In this section, we examine the variation across family and nonfamily firms of different sizes.

We start by sorting active limited liability firms into 10 deciles based on their revenues in 2023 (where 1 represents the smallest and 10 the largest).

We present the proportion of family firms in the total number of firms, employment, revenues, and assets in each decile. Family firms represent the majority of firms, employment, and revenues in all but the largest decile, and a majority of assets in all but the top two deciles:



In the next step, we present financial indicators (means and medians) for family firms (in blue) and nonfamily firms (in red) in each size decile.

For indicators which are levels rather than ratios (e.g. sales, total assets, value added) the numbers are in 2023 kroner (i.e. they are inflation adjusted).

Financial ratios are winsorized at 2.5% and 97.5%.

The skewed distribution of sales is confirmed again in this analysis. The mean and median sales are much higher for firms in the top decile. That is also the decile with the largest gap between family and nonfamily firms.





The total assets for each sales decile confirm the same skewed distribution, as well as the larger size of nonfamily firms in the top decile.





The skewed distribution is also confirmed if we look at the value added (the sum of salaries and earnings generated by the firm).





The number of employees is also significantly higher in the top sales decile. The mean and median number of employees are below 10 for all the other deciles.



The value added (sum of earnings and salaries) per employee is increasing in firm size, and it is higher for family firms in the middle deciles.





Larger firms are also older on average. The relationship between size and age is closer for nonfamily than for family firms.





Sales growth is negative in the smallest decile: the firms with the lowest sales in 2023 had lower sales compared to the previous year. The median growth rates increase across the size deciles, but the relationship between size and mean growth rates is more complex. Family firms usually have slightly lower growth rates, with the notable exception of mean growth rates in the top size decile.



The pattern of asset growth is similar. Median asset growth rates are higher for larger firms. There is evidence of higher growth for large family firms. The distribution of growth rates is also positively skewed, with mean growth rates much higher than median growth rates.





The mean employment growth rates were negative in 2023, especially for small firms. The relationship with firm size is more pronounced for nonfamily firms.



Investment rates (measured as the increase in fixed assets adjusted for depreciation and writedowns) are unrelated to firm size. They are highest among he largest and the smallest firms.





The volatility of sales, measured as their coefficient of variation over the previous three years, is decreasing with firm size for both family and nonfamily firms.





Larger firms are more capital intensive compared to small firms, and nonfamily firms also tend to be more capital intensive than family firms. Those patterns are illustrated by the ratio of firm assets to employees.





The sales generated per unit of assets are increasing in firm size, and usually higher for family firms.





The sales generated per employee increase in firm size:



## Small and medium firms have a higher proportion of cash on their balance sheets. The proportion is also higher for family than for nonfamily firms.





The mean proportion of working capital is increasing in firm size, but the medians do not confirm that monotonicity. The ratio is usually larger for family than nonfamily firms.





The current ratio is larger for small firms, and usually slightly higher for family firms.





#### The accounts receivable, inventory and accounts payable days are all decreasing in firm size:





The ratio of direct costs (costs of goods sold, COGS) to sales is higher in larger firms, especially family firms:





Profitability measured as the operating return on assets (operating earnings after tax divided by total assets) increases in firm size, except for the top deciles, and is usually higher for family than nonfamily firms<sup>16</sup>.



<sup>&</sup>lt;sup>16</sup> The higher profitability of family firms has been documented by multiple studies using listed firms. Similar to us, Andersson et al. (2017) find using Swedish data that family firms are more profitable than nonfamily firms when looking at all limited liability firms.



The return on equity (net income divided by book equity) is also increasing with firm size, except for the top three deciles. It is also usually higher for family than for nonfamily firms.





The return on invested capital is also increasing in firm size, especially in the first deciles, and higher for family firms.





#### The earnings generated by firms per employee are also generally increasing in firm size.





The importance of labor costs, measured as the ratio of total salaries to sales, is highest for mid-size firms, especially family firms:





The average salary per employee is increasing in firm size, especially for family firms:





The mean leverage ratio is gradually decreasing with firm size, while medians are more stable. The distribution of leverage ratios is positively skewed for the smallest firms. Family firms tend to have lower leverage ratios regardless of firm size.





Small firms have relatively more cash on their balance sheets, and that complicates the relationship between net leverage (liabilities minus cash divided by total assets minus cash) and firm size. Family firms have lower net leverage than nonfamily firms.





In terms of the structure of financial debt, larger firms have a higher proportion of short-term to long-term debt.




The interest coverage ratio is increasing in firm size, consistent with decreasing leverage ratios, and is usually higher for family than for nonfamily firms.





Interest expenses are relatively small compared to sales. Among the small firms, many of them have zero financial leverage and zero interest expenses, while some have very high leverage and significant interest expenses relative to their revenues.





Looking at the payments made to shareholders, the dividend payout ratio and the proportion of firms paying dividends is increasing in firm size.





## 9. Financial indicators over time, family and nonfamily firms

The financing, growth, liquidity and profitability of firms can change along the business cycle. The graphs below present financial indicators (means and medians) for every year between 2000 and 2023 for all Norwegian limited liability family firms (in blue) and nonfamily firms (in red).

The sales, total assets, and employment of nonfamily firms are consistently larger than those of family firms through the business cycle. The difference is larger for means than for medians, due to the existence of some very large nonfamily firms. The impact of the 2015 oil shock is visible in sales and assets.





The mean value added is higher in the larger nonfamily firms, and the difference is stable over time. The median value added is declining over time for both family and nonfamily firms.



The median firm age is quite stable over time for both family and nonfamily firms. Family firms typically have a higher average age, but the mean is lowered after 2012 by the new small firms appearing as a result of the 2012 regulatory changes, some of them previous sole proprietorships.



Growth rates follow the business cycle, with a decrease in revenues for instance during the financial crisis of 2008. In general, the means are much higher than the medians, due to the influence of a minority of firms with very large growth rates. The growth rates of nonfamily firms are consistently slightly higher than the growth rates of family firms. (The growth rates for employment between 2014 and 2015 are not representative given the change in the reporting rules.)







The higher growth rates for nonfamily firms are associated with higher investment rates throughout the business cycle.





The volatility of revenues is slightly higher for family firms, and the gap has increased in recent years with the increase in the number of small family-controlled limited liability businesses:

Nonfamily firms are more capital intensive, as illustrated by the higher asset to employee ratio in all years. They also have a higher average value added per employee.





The earnings per employee are quite volatile over time.



The ratio of sales to assets was strongly affected by the 2008 financial crisis and the 2020 Covid-19 crisis for both family and nonfamily firms.



Nonfamily firms have higher average sales per employee throughout the business cycle, where the difference is mostly driven by the largest nonfamily firms.



Since family firms are more labor intensive than nonfamily firms, they have a higher ratio of salaries to sales.



The average salary per employee was higher in nonfamily than in family firms in all years. The caveat is that the difference is driven by large firms, where the missing number of foreign employees may be an issue.



Family firms have higher average liquidity compared to nonfamily firms, as shown by the cash to assets and the current ratio. The increase in the cash to assets ratio after 2005 is partly explained by the higher taxation of dividends, which has increased the cost of payouts to shareholders.



The accounts payable, accounts receivable, and inventory days are fairly stable through the whole period. The first two are higher for nonfamily firms, and the last one slightly higher for family firms.







Related to the liquidity position, family firms have had a higher ratio of working capital to total assets following the 2006 tax reform.



The leverage ratio of family firms is lower than that of nonfamily firms. It has been declining slightly in recent years. Given the increase in cash holdings following the 2006 tax reform, there is a steep decline in net leverage.



The mean proportion of short- to long-term debt has been stable and quite similar for family and nonfamily firms. The median has been declining, while being higher for family than for nonfamily firms.





The average interest coverage ratio is fairly high for both family and nonfamily firms. The coverage ratio has been increasing over time, while leverage and the ratio of interest to sales have been declining.

The ratio of direct costs (cost of goods sold, COGS) to sales is higher for nonfamily than for family firms, and it has been declining in recent years.



The profitability of firms follows the business cycle – it was affected for instance by the financial crisis or the Covid-19 shock. The high ratios around 2004, especially for returns on equity, are explained by the effects of the 2006 tax reform, which induced firms to pay very high dividends and thus reduce their equity prior to the tax increase.



Family firms are usually more profitable than nonfamily firms across the business cycle.



The dividend payout ratio and the proportion of dividend-paying firms were both very high prior to the 2006 tax reform. Since then they have stabilized at a lower lever for both family and nonfamily firms.





## 9.2. Financial indicators over time, large family and nonfamily firms

While family firms are smaller than nonfamily firms on average, there are nevertheless many large family firms that are of comparable size to nonfamily firms. The graphs below present financial indicators (means and medians) for every year between 2000 and 2023 for all large Norwegian limited liability family firms (in blue) and nonfamily firms (in red). Large firms are defined as firms with at least 100 employees and 100 million (2023) kroner in sales.

Number of firms 20.2 Number of family firms • Number of nonfamily firms

The number of large nonfamily firms increased gradually from around 800 to around 1,200. The number of large family firms was between 400 and 600 during the same period.

Large nonfamily firms are larger on average than large family firms. The difference is larger in means than medians, due to the influence of a few very large nonfamily firms.



Large family firms are slightly older than large nonfamily firms.



Family firms are less capital intensive than family firms, and that is reflected in lower assets per employee and value added per employee ratios.





Investment rates are similar for family and nonfamily firms. Family firms have more volatile revenues.





Similar to their smaller counterparts, large family firms have more liquidity than large nonfamily firms, as shown by their higher cash to assets ratio and current ratio.





## Family firms have a higher ratio of direct costs (COGS, cost of goods sold) to sales.

Family firms tend to be more profitable than nonfamily firms, at least as measured by the return on assets and return on equity.











Leverage ratios are similar for family and nonfamily firms. The leverage ratios of family firms declined at a slightly faster rate following the 2006 tax reform. Nonfamily firms have a higher share of short-term financial debt, and they have lower interest costs, measured by the interest coverage ratio and the interest to sales ratio.











Family firms paid higher dividends prior to the 2006 tax reform. Following that, family and nonfamily firms have similar dividend payout ratios, but family firms are more frequent payers.





## 10. Financial indicators for family and nonfamily firms by industry, 2000-2023

The graphs below present financial indicators (means and medians for the 2000-2023 period) by industry for family firms (in blue) and nonfamily firms (in red)<sup>17</sup>.

Nonfamily firms are larger than family firms in terms of sales, total assets, number of employees, and value added in all industries. The difference is larger when we look at means than when we look at medians, due to the existence of a subset of very large nonfamily firms.



<sup>&</sup>lt;sup>17</sup> This section presents averages (means and medians) by industry for the entire 2000-2023 period. The numbers for just 2023 can be found in the Appendix.













The typical firm age of family and nonfamily firms is quite similar across industries.



Nonfamily firms are more capital intensive than family firms. The difference is particularly large in mining and oil and utilities.



The lower capital intensity of family firms is also illustrated by their higher sales to assets ratios compared to sales and value added per employee.





The growth rates of sales, assets, and employment have been higher form nonfamily than for family firms in most industries. An exception is multisector firms, which are active in multiple industries.




The mean investment rates of family firms is comparable to that of nonfamily firms. The highest investment rates in recent years have been in IT and mining and oil.



The volatility of sales is similar or slightly higher in the case of family firms.



The liquidity position of family firms is usually better across industries, as illustrated by the cash to assets ratio and the current ratio.







The ratio of direct costs (COGS) to sales is slightly lower in family firms than in nonfamily firms across industries. The largest ratio, unsurprisingly, is in retail and wholesale.



Family firms are more profitable than nonfamily firms across industries, as measured by operating return on assets, returns on equity, and returns on invested capital.







Given their higher labor intensity, family firms often generate lower earnings per employee – with a large difference in mining and oil. The ratio of salaries to sales is comparable for family and nonfamily firms, and is highest in services, including public services. Consistent with the existing literature (e.g. Sraer and Thesmar 2007), salaries per employee seem to be slightly lower in family firms.





The leverage ratios of family and nonfamily firms are similar across industries, while the net leverage is lower, especially in services, IT, and mining and oil.









Average dividend payout ratios and the proportion of dividend payers have been quite similar across industries, and between family and nonfamily firms.





# 11. The governance of limited liability Norwegian firms

### 11.2. Ownership

We begin by examining the ownership of Norwegian limited liability firms. We group owners in families, look for ultimate ownership, and group firms in business groups so that one business group is one observation in our dataset. For business groups, we choose the holding company with the largest and most representative board.

The proportion of single-owner firms has increased over time, partly as a result of the increase in the number of small limited liability firms after 2012. Around three quarters of limited liability firms are now completely owned by one individual, family, financial institution or foreign entity.



The median limited liability firm was a single-owner firm throughout the sample period. The mean ownership share of the largest shareholder has been increasing slightly over time from 80% to around 90%.





As the ownership share of the largest shareholder has been increasing, the share of secondlargest shareholders has been decreasing.

The average ownership share of insiders (CEO and members of the board of directors) in limited liability firms has always been high, but there has been an additional gradual increase in recent years.



# 11.3. The board of directors

The typical board size in Norway is small by international standards. Given the increase in the number of small firms in recent years there has been an additional decrease in the average board size.



The directors of Norwegian companies are elected for limited terms – typically 2 years – and can be replaced by shareholders anytime. The board turnover rate (the proportion of directors replaced in a given year) over time reflects important events, such as the 2004 board gender quota for ASA firms.



On average, directors own 80% of the equity in limited liability firms, and the chair of the board owns around 60%.





#### The proportion of women on board is around 25%.





The average age of board members (both men and women) and board chairs has gradually increased in recent years.







The standard deviation of director ages has been fairly stable over time.



# 11.4. CEOs

In line with the trend towards higher ownership concentration, the CEO ownership share and the proportion of firms where the CEO controls a majority of the shares has been increasing over time.





The proportion of female CEOs is low, but it has been gradually increasing from 14% to 18%:





The average CEO age has been increasing over time regardless of gender. Female CEOs are still slightly younger than male CEOs.





# 12. Governance: Family and nonfamily firms

#### 12.2. Ownership

The ownership concentration of family firms is higher than that of nonfamily firms. Moreover, among family firms, the proportion of firms where the family owns all the shares has increased from around 70% to around 90%. In parallel, the average size of the second-largest stake in the firm has been decreasing.







Insiders (directors and CEOs) typically own a large proportion of the firm's equity. This indicates that large shareholders are usually also represented on the board in nonfamily firms.



### 12.3. The board of directors

Family firms have on average smaller boards compared to nonfamily firms. There has been a parallel downward trend in the mean board size for family and nonfamily firms.



Board turnover rates are higher for nonfamily firms than for family firms.



The mean ownership share of the chair is around 20% in nonfamily firms – and a much higher 80% in family firms. In both family and nonfamily firms, directors hold on average a majority of the shares.





The proportion of women on board is higher in family than in nonfamily firms.



The average age of directors and chair has gradually increased over time, and in recent years the average age of nonfamily firm directors is higher in nonfamily than in family firms, driven mainly by male directors. The median standard deviation of director ages for family firms is much lower than the mean, pointing to the existence of a set of boards with directors from different generations.











# 12.4. CEOs

The typical CEO ownership share is much larger in family firms. The typical age of CEOs is slightly lower than that of directors, and it has been increasing slightly over time, especially in nonfamily firms.





The proportion of female CEOs has been gradually increasing over time, and it is higher in family firms (around 20%) compared to nonfamily firms (around 15%). Female CEOs tend to be younger on average than male CEOs.







# 13. Governance and firm size

Small firms tend to have more concentrated ownership, fewer owners, and controlling CEOs. Family firms also tend to have these features, and they also tend to be smaller than nonfamily firms. Some of the governance characteristics of family firms identified in previous sections may therefore be due to their size rather than their ownership. It may therefore be useful to look at family and nonfamily firms of various sizes to separate the influence of size from that of ownership.

In the following analysis, we classify firms into deciles based on their 2023 revenues. We then present governance characteristics for family and nonfamily firms in each size decile.

Four out of five firms are family firms in the smallest 6 size deciles are family firms. The proportion is gradually decreasing in larger size deciles, but family firms still form a majority except the largest decile, where they still represent 40%.



# 13.2. Ownership

Among family firms, around 90% are single-owner firms except in the top two size deciles. Even among family firms that are in the top size decile, three quarters have 100% of their shares owned by the family.



The number of owners increased with firm size. However, within each size decile, family firms have significantly fewer owners. In the top size decile, nonfamily firms have 45 ultimate owners on average, while family firms have 5.



The ownership share of the second-largest shareholder increases with firm size in family firms, and it decreases for nonfamily firms. For family firms, this is linked to the decreasing average share of the controlling family, while for nonfamily firms the average ownership share in the firm in decreasing.



The ownership share of insiders (CEOs and directors) is higher for family than for nonfamily firms in all size deciles, but it is quite high across the board. Even in the largest nonfamily firms, insiders have on average a two-thirds majority.



# 13.3. The board of directors

The number of directors on the board is increasing with firm size, but it is always higher for nonfamily than for family firms.



The proportion of women on board is lower in the smallest and the largest firms. It is also higher in family than in nonfamily firms across the size distribution.





On average, female directors are younger than male directors, and the difference is highest is among the largest firms.



The average director age is similar in family and nonfamily firms, except for the smallest firms, where family firm directors tend to be older. Female directors in family firms tend to be older than female directors in nonfamily firms, especially in the largest and smallest firms.






The age of the chair of the board is higher in nonfamily than in family firms, except in the smallest firms.



The ownership share of the chair is much larger in family than in nonfamily firms across the size distribution.



The ownership share of directors is much larger than just the chair's holdings in nonfamily firms. Directors in family firm own a higher share compared to directors in nonfamily firms, and the difference is the largest in the top size decile.



In family firms, the chair of the board is a family member in most cases. The proportion is only slightly lower among the largest firms.



Apart from monitoring the management and advising the firm on its strategy, the board can also serve to coordinate and manage potential conflicts between shareholders (Burkart et al. 2023). It is therefore interesting to examine the board participation of minority shareholders. Nonfamily shareholders are more likely to be represented on the board in larger family firms.



The average proportion of directors who are family members is above three quarters in all size deciles. It is lowest in the top decile, where minority shareholders are more likely to be present on the board.



In a majority of family firms, all board members are family members. That applies across the size distribution – the proportion is above 60% even in the largest size decile.



### 13.4. CEOs

The CEOs of nonfamily firms are slightly older than the CEOs of family firms, but the difference is small. The CEOs of small and medium enterprises tend to be slightly older on average.



The proportion of female CEOs is around 20% across most size deciles – but it is lower in the top decile.



Female CEOs tend to be younger than male CEOs across the size distribution, with a larger difference in small firms. There do not seem to be systematic differences between family and nonfamily firm CEOs.







The proportion of female CEOs is higher among family firms in the smallest 5 size deciles, and quite similar in the larger ones.



The ownership share of the CEO is much larger in family than in nonfamily firms across the size distribution. It is only slightly declining with firm size.



At least four out of five family firms have a family CEO across the size distribution. The proportion is relatively lower among the smallest and the largest firms.



In a majority of family firms, the family CEO is the controlling shareholder. The proportion is lowest (74%) in the top size decile.



Indeed, the family has both the chair and the CEO position in more than two thirds of family firms across the size distribution. The proportion is slightly smaller among tha smallest and largest firms.



## 14. Governance: Large family and nonfamily firms

In this section, we take a closer look at large family and nonfamily firms. Large firms are defined as firms with at least 100 employees and 100 million (2023) kroner in sales.

#### 14.2. Ownership

Ownership concentration is high even among large family firms. The proportion of single-owner firms is significantly higher among large family firms than among large nonfamily firms, and the proportion has slightly increased in recent years to around 80%.



The ownership share of the largest owner (individual or family) in family firms is around 90% on average.



The mean share of the second-largest shareholder in large family firms is at around 15% and it has declined slightly over time.



The ownership share of insiders (CEOs and directors) is above two-thirds in both large family and large nonfamily firms. It is still significantly higher for large nonfamily firms.



#### 14.3. The board of directors

The typical board size of large family firms is smaller than that of large nonfamily firms.



Board turnover is lower in large family firms than in large nonfamily firms.



The ownership share of chairs and directors is much higher in large family firms than in large nonfamily firms.





The proportion of women on board is higher – at above 25% - in large family firms than in large nonfamily firms, although the gap has narrowed somewhat in recent years..



The typical ages of chairs and directors are very similar in large family and nonfamily firms. There has been a trend towards older directors in recent years.









The standard deviation of director ages has been quite stable over time. The larger difference between the mean and the median for family firms comes from smaller boards and boards where different generations in the family are present on the board.



Looking at just large family firms, the proportion of firms where the family has chair position has been fairly stable over time at above 80%.



Family members sit on the board in around 95% of the cases, and on average four out of five directors are family members. The proportion of boards where the family has all board seats has increased from one half to more than two thirds.









Minority shareholders are represented on the board in 10% of large family firm boards.

#### 14.4. CEOs

The ownership stake of the CEO is larger for family than for nonfamily firms. In most large family firms, the CEO controls 100% of the equity.



The average age of the CEO is very similar for family and nonfamily firms, and it has gradually increased over time.







The proportion of female CEOs in large family and nonfamily firms has converged in recent years at around 12%.







Looking at just the large family firms, the CEO is a family member in 9 out of 10 cases, and in 4 out of 5 the family has both the chair and the CEO position. The proportion of firms where the CEO controls a majority of the shares has increased from around two thirds to around three quarters.







# 15. Family firms with/without minority investors and family CEOs

Large majorities of family firms have the family as the only shareholder and a CEO who is also a family member. It is therefore interesting to see whether family firms with nonfamily shareholders and nonfamily CEOs are different on important dimensions.

The graphs below present financial indicators for 6 subgroups of family firms:

- Firms which are entirely owned by their CEO (no minority, single-owner)
- Firms which are entirely owned by the family, have multiple family owners, and a family CEO (no minority, family CEO)
- Firms which are entirely owned by the family, have multiple family owners, and a nonfamily CEO (no minority, nonfamily CEO)
- Firms which are controlled by their CEO and which also have owners who are unrelated to the CEO (minority, single family owner is CEO)
- Family firms with multiple family owners which also have nonfamily owners, and have a CEO that belongs to the controlling family (minority, family CEO)
- Family firms which also have nonfamily owners, and have a nonfamily CEO (minority, nonfamily CEO)


























































































































## 16. Foreign ownership

In this section, we present statistics on foreign-controlled firms active in Norway in 2023. We define foreign-majority (foreign-controlled) firms as firms or business groups where foreign individuals or entities ultimately control at least 50% of the equity.

The proportion of firms with foreign majority ownership is highest in mining and oil and IT. It is lower in fishing and farming.



In terms of revenues, foreign-controlled firms are important in services, mining and oil, and IT.



Foreign-controlled firms represent a significant proportion of assets in services, IT, and mining and oil, and very small percentage in fishing, forestry, public services, and utilities.



In terms of employment, foreign-controlled firms represent the largest proportion in IT, followed by mining and oil and real estate.



Foreign-controlled firms are typically smaller than locally-controlled firms:







Firms with foreign controlling owners have similar average ages to local firms across industries.



## 17. Corporate and personal finance

In the case of family firms, the personal finances of the controlling family are closely linked to the firm. Shocks to the firm affect the family, and personal financial shocks to the family propagate to the firm. An additional link is created by the fact that the firm is part of the family's taxable wealth.

## 17.2. Wealth taxation over time

The wealth tax has been the focus of public policy debate in Norway for many years. It has also been part of international debates on tax systems and inequality (Zucman 2023).

The distribution of wealth is skewed, and as a result the distribution of wealth tax payments is also highly positively skewed.

The graphs below illustrate this property of the distribution. The first graph presents the wealth tax payments of families that control family firms in 2021 (in million kroner, adjusted for inflation to 2023). The second graph presents the distribution of the natural logarithm of 1+the amount of wealth tax payment in kroner, for the same sample and year.





In this section we describe the evolution of wealth taxation for business owners over time. To address the issue of skewness, we group family firms in 8 broad bands based on the net (taxable) wealth of their controlling owners.

We form the following groups of family businesses based on the net wealth of the controlling family:

- Group 0: negative net wealth;
- Group 1: net wealth between 0 and 500,000 kroner (adjusted for inflation, in 2023 kroner);
- Group 2: net wealth between 500,000 and 1,000,000 kroner;
- Group 3: net wealth between 1,000,000 and 5,000,000 kroner;
- Group 4: net wealth between 5,000,000 and 10,000,000 kroner;
- Group 5: net wealth between 10,000,000 and 50,000,000 kroner;
- Group 6: net wealth between 50,000,000 and 100,000,000 kroner;
- Group 7: net wealth above 100,000,000 kroner.

The graph below presents the number of firms in each group and year:



The proportion of family firms in each group is fairly stable over time:



Around one third of business owners have negative net wealth, where the net wealth is calculated as the difference between gross wealth (the family's total assets) and personal liabilities. That means they do not pay the wealth tax. Another large group is business owners with a net wealth between 1 and 5 million kroner (group 3 in our classification). The top two groups, with net wealth above 100 million kroner or between 50 and 100 million kroner (groups 6 and 7) stand for between 1 and 3% of business owners each depending on the year.

As it may be expected, business owners with higher net wealth pay larger amounts of wealth tax. The graph below presents the mean amount paid by business owners in each group, in million (2023, adjusted for inflation) kroner.



Since the distribution of tax payments is highly skewed, it is easier to see the evolution of tax payments for groups 0-6 if we exclude group 7 (with the largest tax payments):



To complete the picture, we can also look at the evolution of gross wealth (total personal assets) for each group of business owners over time. There is obviously a positive correlation between the net and the gross wealth, and the ranking of the groups is the same using the gross wealth:





Wealth tax payments represent a cash outflow for the tax payers. It is therefore interesting to compare the wealth tax payments of the business owners to their personal liquid assets, which include bank savings and listed securities.

The ratio of tax payments to liquid assets has some extreme outliers in group 7. To make it easier to follow the evolution of the ratio over time and reduce the influence of extreme outliers, we winsorize the ratio of wealth tax payments at 97.5% every year. We can see that wealthier business owners also pay more in the wealth tax relative to their personal liquid assets. The ratio has increased in the second half of our time period.



A similar relationship can be seen if we focus on the median ratio for each group:



## 17.3. Wealth taxation and firm characteristics

In this section, we link the business owners in each group of net wealth to the characteristics of the firms they control.



Wealthier business owners tend to have relatively older businesses.

Wealthier business owners also have larger businesses, whether we measure that by sales, total assets, or employment. This is to be expected, since the net value of the firm's assets is part of the owner's tax base:









A similar picture emerges if we look at value added (calculated as the sum of earnings – the revenues going to capital – and salaries – the earnings going to labour):





Firms with wealthier owners tend to be more capital intensive, as indicated by the ratio of total assets to employees:



A similar picture emerges if we look at sales or value added per employee:









The earnings (net income) generated per employee are also correlated with firm size and the net wealth of the owners. For the top groups, we can also observe a significant cyclical pattern, with downturns in 2008 and 2015:



The various groups are much closer when we look at growth rates, where there seems to be no clear hierarchy:







The differences in risk, measured as the coefficient of variation for sales over a three-year period, are small, with the largest and smallest firms (most/least wealthy owners) having slightly higher volatility:



The higher capital intensity of larger firms with wealthier owners is also illustrated by their higher ratio of working capital to total assets, and their lower ratios of salaries and sales to assets.






Looking at the firms' liquidity, larger firms with wealthier owners have higher current ratios, but they also have a lower proportion of cash on their balance sheet.





There is little difference across the groups in terms of their accounts payable, and accounts receivable and inventory days, and all groups show an upward trend:







Looking at financing, leverage varies monotonically across our groups, and differences are stable over time. Smaller firms with less wealthy owners are more highly levered.



When we look at net leverage (total liabilities minus cash divided by total assets minus cash), the picture is somewhat different. Medium-sized firms have low and decreasing net leverage as a result of their increasing cash holdings.



The ratio of short-term to long-term debt does not show a very clear relationship, but in general larger firms have a larger proportion of long-term debt compared to smaller firms.



Interest expenses relative to sales are highest for large firms and wealthy owners. The interest coverage ratio has been increasing in recent years for all groups.





We can also compare profitability across the groups. The ratio of direct costs (the cost of goods sold) to sales has been decreasing over time except for the top two groups, where it has been stable at a relatively lower level.



The return on assets (ROA, calculated as the ratio of the operating income net of tax to total assets or the ratio of the net income plus interest to total assets) varies with the business cycle. It is lower for small firms, and larger for large and medium-sized firms.





The return on equity (ROE, calculated as the ratio of net income to book equity) varies both with the business cycle and as a result of the 2006 dividend tax reform, where the large dividends prior to the tax increase temporarily reduced book equity. It is lower for large firms with wealthy owners.







The dividend policy of Norwegian firms was strongly affected by the 2006 tax reform. Following the reform, large firms with wealthy owners have higher payout ratios and they pay dividends more often.







The differences in the investment rate (relative change in fixed assets adjusted for depreciation and write-downs) across the groups are small. Smaller, younger firms have slightly higher rates.



# 18. Location, firms and owners

### 18.1. Centrality and family firms

Ensuring the vitality of economic activity in outlying areas has long been a policy priority in Norway. Statistics Norway provides a centrality index

(https://www.ssb.no/befolkning/folketall/artikler/sentralitetsindeksen) that classifies municipalities in Norway into 6 categories, ranging from the most central (index value 1, the core of the Oslo region), through the other large cities (index value 2) to the least central (index value 6). For instance, Oslo, Bærum and Lillestrøm are in centrality class 1, Bergen, Stavanger, Trondheim, Asker, Tønsberg, Moss, Hamar, Drammen are in class 2, Kristiansand, Gjøvik, Larvik, Sandefjord, Ålesund, Lillehammer, Tromsø, Bodø are in class 3, Alta, Molde, Gol, Voss, Hvaler are in class 4, Lærdal, Årdal, Hol, Røros, Vadsø are in class 5, Nordkapp, Frøya, Dovre, Ulvik are in class 6.

In this section we present statistics on the relative importance and the characteristics of family and nonfamily firms from regions with different centrality. We attach to every firm the centrality index of the location where it has its registered headquarters. In the case of business groups, we choose the location of the entity where it consolidates its financial reporting. All the statistics are for the year 2023.



Family firms represent a large majority in all regions, but their proportion is higher outside the large cities (regions 1 and 2).

Left scale: the number of firms, right scale: the proportion of family firms.



Family firms also represent a majority of employment outside the large cities.

Left scale: the number of firms, right scale: the proportion of employees in family firms.

Almost half of the employees in nonfamily firms work in firms headquartered in the Oslo region. Employment in family firms is much more evenly distributed across regions. A majority of employees in family firms work in firms headquartered outside the large cities.



The typical number of employees in a nonfamily firm decreases as we move to less central regions, while average employment in family firms is stable.



Typical salaries are quite close for family and nonfamily firms outside the Oslo region.



Family firms represent just 10% of revenues of firms headquartered in the Oslo region. Outside the large cities, however, family firms stand for close to 40%.



Left scale: the number of firms, right scale: the proportion of revenues generated by family firms.

Most nonfamily firm revenues are concentrated in firms headquartered in the Oslo region – while family firm revenues are much more evenly distributed.



The gap between the mean revenues of family and nonfamily firms is also much smaller outside the large cities.





The pattern of family firms being more important outside the large cities is confirmed when we look at assets. Family firms represent just 10% of the assets of firms headquartered in the Oslo region, but around one third for firms headquartered outside the large cities.



Left scale: the number of firms, right scale: the proportion of assets in family firms.

More than two-thirds of the assets of nonfamily firms are in firms headquartered in Oslo, while the proportion for family firms is just above one third.



The gap between the size of family and nonfamily firms is smaller outside the large cities. The size distribution of nonfamily firms is also highly skewed.







### On average, firms in less central regions are slightly older.

### The profitability of family firms is relatively higher in less central regions.







233



Looking at financing and payout policy, the net leverage of family firms is higher in less central areas, and the dividend payout ratio is lower in less central areas for both family and nonfamily firms.





Ownership is slightly more concentrated in more central regions. Conversely, the share of the second-largest owners and the total number of ultimate owners increases in less central regions.







Family participation in the governance of family firms is quite strong across regions. Firms in least central regions are slightly less likely to have both family CEOs and family chairs of the board.



In line with the more concentrated ownership, a higher proportion of firms in more central regions have CEOs who are also the controlling owners.



Family firms in less central areas have a slightly lower proportion of family members on the board, and a higher representation of minority shareholders.







The average board size is larger, and directors tend to be slightly older in less central regions.



The proportion of women on board is higher in family firms across all regions. For nonfamily firms, the proportion is increasing as we move to less central regions.



The age heterogeneity of directors is higher in less central regions.



Board composition is more stable in family firms across all regions.



Similar to directors, CEOs in less central regions tend to be slightly older. The proportion of female CEOs is higher in family firms, although the difference is smaller in less central areas.





### 18.2. Fish farming: an industry for less central regions

Fish farming is an example of an industry that is strongly represented in less central areas. The graphs below show that a majority of fish farming firms are headquartered in centrality regions 5 and 6. The highest proportion of revenues come from firms headquartered in regions 6, 2, and 5 (Mowi is headquartered in Bergen – centrality index 2, and Salmar is headquartered in Frøya – centrality index 6). The same regions dominate in terms of total assets, employment, and value added.











Fish farming firms in less central areas are larger on average, as shown by their sales, assets, employment, and value added. (Mean numbers for region 2 are largely driven by Mowi. As usual, size distributions are skewed, with a large difference between means and medians.)





Consistent with having larger, more mature firms, profitability is also higher for firms in less central areas.



Family firms represent a large proportion across all regions.



### Looking at trends over time, the average firm size has been relatively stable.







If we look at the total sales, assets, employment, and value added by fish farming firms during the same period, we can see an upward trend, especially in more recent years, with a doubling



or near-doubling for all indicators. The year 2023, the first year after the introduction of the resource tax in fish farming, is marked by a small decrease.

The average profitability of fish farming firms was high around the time of the exchange rate shock of 2014-2015, and lower in more recent years.



### Other financial indicators have been fairly stable over time.







## 18.3. Centrality and foreign ownership



Firms controlled by non-Norwegian individuals or corporations represent a relatively higher proportion in more central regions.

Foreign-controlled firms tend to be larger and to represent a larger proportion of revenues in more central regions.





A similar picture emerges if we look at the assets of foreign- and locally-controlled firms.




Foreign-controlled firms represent a small, but significant proportion of the employment of firms headquartered in central regions. In contrast, they stand for less than 1% in the three least central regions.





Two thirds of foreign-controlled firms are headquartered in large cities (centrality regions 1 and 2), compared to less than one half for locally controlled firms.



The distribution of sales, assets, and employment is even more concentrated in firms headquartered in large cities (centrality 1 and 2).







## References

Adams, R.B., B.E. Hermalin, and M.S. Weisbach, 2010. The role of boards of directors in corporate governance: A conceptual framework and survey. *Journal of Economic Literature* 48 (1), 58–107.

Admati, A., P. Pfeiderer, and J. Zechner, 1994. Large shareholder activism, risk-sharing, and financial market equilibrium, *Journal of Political Economy* 102, 1097-1130.

Allen, F., E. Carletti, and R. Marquez, 2009. Stakeholder capitalism, corporate governance and firm value. Working paper, https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=968141.

Almeida, H., and D. Wolfenzon, 2006. Should business groups be dismantled? The equilibrium costs of efficient internal capital markets. *Journal of Financial Economics* 79, 99–144.

Amit, R., and B. Villalonga, 2014. Financial performance of family firms. In L. Melin, M. Nordqvist and P. Sharma (Eds.), *The SAGE Handbook of Family Business*. London: Sage.

Andersen, S., and Kasper Meisner Nielsen, 2012, Ability or Finances as Constraints on Entrepreneurship? Evidence from Survival Rates in a Natural Experiment, *Review of Financial Studies* 25, 3684-3710.

Anderson, R., and D.M. Reeb, 2003. Founding-family ownership and firm performance: Evidence from the S&P 500. *Journal of Finance* 58, 1301–1328.

Anderson, R., S.A. Mansi, and D.M. Reeb, 2003. Founding-family ownership and the agency cost of debt. *Journal of Financial Economics* 68, 263–285.

Andersson, F.W., D. Johansson, J. Karlsson, M. Lodefalk, and A. Poldahl, 2017. The characteristics and performance of family firms: Exploiting information on ownership, kinship, and governance using total population data. *Small Business Economics* 49, 1–18.

Astrachan, J.H., and M.C. Shanker, 2003. Family businesses' contribution to the US economy: A closer look. *Family Business Review* 16, 211–219.

Audretsch, D. B., 2018, Entrepreneurship, economic growth, and geography, *Oxford Review of Economic Policy* 34, 637–651.

Barontini, R., and L. Caprio, 2006. The effect of family control on firm value and performance: Evidence from continental Europe. *European Financial Management* 12, 689–723.

Becht, M., P. Bolton, and A. Roëll, 2003. Corporate governance and control. In G. Constantinides, M. Harris, and R. Stulz (Eds.), *Handbook of the Economics of Finance*, Volume 1A, 1-109. Amsterdam: North-Holland.

Bena, J., M. Ferreira, P. Matos, P. Pires, 2017. Are foreign investors locusts? The long-term effects of foreign institutional ownership, *Journal of Financial Economics* 126, 122-146.

Bennedsen, M., F. Pérez-González, K. Nielsen, and D. Wolfenzon, 2007. Inside the family firm: The role of families in succession decisions and performance. *Quarterly Journal of Economics* 122, 647–691. Bennedsen, M., F. Pérez-González, and D. Wolfenzon, 2010. The governance of family firms, In Baker, H. K., and R. Anderson (Eds.), *Corporate Governance: A Synthesis of Theory, Research, and Practice*, Hoboken, NJ: Wiley.

Berglann, H., E.R. Moen, K. Røed, Jens Fredrik Skogstrøm, 2011, Entrepreneurship: Origins and returns, *Labour Economics* 18, 180–193.

Berk, J., R.C. Green, and V. Naik, 1999. Optimal investment, growth options, and risk, *Journal of Finance* 54, 1553–1607.

Berle, A.A., and G.C. Means, 1932. *The Modern Corporation and Private Property*. New York: Macmillan.

Berzins, J., Ø. Bøhren, and B. Stacescu, 2018a. Shareholder conflicts and dividends. *Review of Finance* 22, 1807–1840.

Berzins, J., Ø. Bøhren, and B. Stacescu, 2018b. The governance and finance of family firms: A close look at the population. CCGR Research Report 1/2018.

Berzins, J., Ø. Bøhren, and B. Stacescu, 2019. Dividends and taxes: The moderating role of agency conflicts. Journal of Corporate Finance 58, 583-604.

Bhide, A., 1993. The hidden costs of stock market liquidity, *Journal of Financial Economics* 34, 31-51.

Bøhren, Ø., and B.A. Ødegaard, 2000. The ownership structure of Norwegian firms: Characteristics of an outlier, Research report 13/2000, BI Norwegian Business School.

Bøhren, Ø., R. Priestley, and B.A. Ødegaard, 2008. Short-termism and the value of the firm, Working paper, BI Norwegian Business School.

Bøhren, Ø., and R.Ø. Strøm, 2010. Governance and politics: Regulating independence and diversity in the board room, *Journal of Business Finance and Accounting* 37, 1281-1307.

Bøhren, Ø., and M. Josefsen, 2013. Stakeholder rights and economic performance: The profitability of non-profits, *Journal of Banking and Finance*, 4073-4086.

Bøhren, Ø., and Siv Staubo, 2016, Mandatory Gender Balance and Board Independence, *European Financial Management* 22, 3-30.

Bøhren, Ø., B. Stacescu, L. Almli, and K.L. Søndergaard, 2019. When does the family govern the family firm? *Journal of Financial and Quantitative Analysis* 54, 2085 – 2117.

Boot, A.W., R. Gopalan, and A.V. Thakor, 2007. Market liquidity, investor participation and managerial autonomy: Why do firms go private?, *Journal of Finance* 63, 2013-2059.

Bozec, Y., and C. Laurin, 2008. Large shareholder entrenchment and performance: Empirical evidence from Canada. *Journal of Business Finance and Accounting* 35, 25–49.

Burkart, M., D. Gromb, and F. Panunzi, 1997. Large shareholders, monitoring, and the value of the firm, *Quarterly Journal of Economics* 112, 693-728.

Burkart, M., S. Miglietta, and C. Ostergaard, 2023, Why Do Boards Exist? Governance Design in the Absence of Corporate Law, *Review of Financial Studies* 36, 1788–1836.

Claessens, S., S. Djankov, and L.H.P. Lang, 2000. The separation of ownership and control in East Asian corporations. *Journal of Financial Economics* 58, 81–112.

Coffee, J.C., 1991. Liquidity versus control: The institutional investor as corporate monitor. *Columbia Law Review* 91, 1277–1368.

Demsetz, H., 1983. The structure of ownership and the theory of the firm. *Journal of Law and Economics* 26, 375–390.

Ahern, K. R. and Amy K. Dittmar, 2012, The Changing Of The Boards: The Impact On Firm Valuation Of Mandated Female Board Representation, *Quarterly Journal of Economics* 127, 137-197.

Doidge, C., G. A. Karolyi, R. M. Stulz, 2017, The U.S. listing gap, *Journal of Financial Economics* 123, 464-487.

Easterbrook, F., 1984. Two agency–cost explanations of dividends. *American Economic Review* 74, 650–659.

Eckbo, B. E. (ed.), 2007. *Handbook of Corporate Finance: Empirical Corporate Finance*, Volume 2. Amsterdam: Elsevier.

Eckbo, <u>B. E., Nygaard</u>, K., Thorburn, <u>K. S.,</u> 2022, Does Mandatory Board Gender-Balancing Reduce Firm Value?, <u>European Corporate Governance Institute - Law Working Paper No.</u> 629/2022

Edmans, A. and C.G. Holderness, 2017. Blockholders: A survey of theory and evidence. In Hermalin, B.E., and M.S. Weisbach, *The Handbook of the Economics of Corporate Governance*, Volume 1. Amsterdam: Elsevier.

European Commission, 2003. Commission Recommendation of 6 May 2003 concerning the definition of micro, small and medium-sized enterprises.

https://publications.europa.eu/en/publication-detail/-/publication/6ca8d655-126b-4a42-ada4-e9058fa45155/language-en.

European Commission, 2009. Final Report of the Expert Group. Overview of Family-Business-Relevant Issues: Research, Networks, Policy Measures and Existing Studies, at <u>http://ec.europa.eu/enterprise/policies/sme/</u>

Faccio, M., L.H.P. Lang, and L. Young, 2001. Dividends and expropriation. *American Economic Review* 91, 54-78.

Fjærli, E., D. Iancu, A. Raknerud, 2013, Facts about entrepreneurship in Norway. Who become entrepreneurs and how do they perform?, SSB Report 52/2013.

Faccio, M., and L.H.P. Lang, 2002. The ultimate ownership of Western European corporations. *Journal of Financial Economics* 65-365–395.

Faccio, M., J.J. McConnell, and D. Stolin, 2006. Returns to acquirers and unlisted targets, *Journal of Financial and Quantitative Analysis* 41, 197-220.

Faccio, M., Maria-Teresa Marchica and Roberto Mura, 2011, Large Shareholder Diversification and Corporate Risk-Taking, *Review of Financial Studies* 24, 3601-3641.

Fagereng, A., L. Guiso, D. Malacrino, L. Pistaferri, 2020, Heterogeneity and Persistence in Returns to Wealth, *Econometrica* 88, 115-170.

Franks, J., C. Mayer, and S. Rossi, 2009. Ownership: Evolution and regulation. *Review of Financial Studies* 22, 4009–4056.

Giannetti, M., and A. Simonov, 2009, Social Interactions and Entrepreneurial Activity, *Journal of Economics & Management Strategy* 18, 665-709.

Giroud, X., 2011. Corporate governance, product market competition, and equity prices, *Journal of Finance* 66, 563–600.

Giroud, X., and H.M. Mueller, 2010. Does corporate governance matter in competitive industries? *Journal of Financial Economics* 95, 312–331.

Gómez-Mejía, L. R., Haynes, K. T., Núñez-Nickel, M., Jacobson, K. J. L., and Moyano-Fuentes, J., 2007, Socioemotional Wealth and Business Risks in Family-controlled Firms: Evidence from Spanish Olive Oil Mills. *Administrative Science Quarterly* 52, 106-137.

Hadlock, C.J., and J.R. Pierce, 2010. New evidence on measuring financial constraints: Moving beyond the KZ index, *Review of Financial Studies* 23, 1909–1940.

Hagen, I.M., 2016. Participation and co-determination: Why some arrangements fail and others prevail. In F. Engelstad and A. Hagelund (Eds.), *Cooperation and Conflict the Nordic Way*, 77–95. Berlin: De Gruyter.

Hall, R. E., and Susan E. Woodward, 2010, The Burden of the Nondiversifiable Risk of Entrepreneurship, *American Economic Review* 100, 1163–1194.

Hamilton, B. H., 2000, Does Entrepreneurship Pay? An Empirical Analysis of the Returns to Self-Employment, *Journal of Political Economy* 108, 604-631.

Hermalin, B.E., and M.S. Weisbach, 2017. *The Handbook of the Economics of Corporate Governance*, Volume 1. Amsterdam: Elsevier.

Hombert, J., A. Schoar, D. Sraer, D. Thesmar, 2020. Can Unemployment Insurance Spur Entrepreneurial Activity? Evidence from France, *Journal of Finance* 75, 1247-1285.

Hope, O.K., and J.C. Langli, 2010. Auditor independence in a private firm and low litigation risk setting. *Accounting Review* 85, 573–605.

Hvide, H. K., and Jarle Møen, 2010, Lean and Hungry or Fat and Content? Entrepreneurs' Wealth and Start-Up Performance, *Management Science* 56, 1242–1258.

Hvide, H. K., and G. A. Panos, 2014, Risk tolerance and entrepreneurship, *Journal of Financial Economics* 111, 200-223.

Hvide, H. K., and P. Oyer, 2017, Who Becomes a Successful Entrepreneur? The Role of Early Industry Exposure, Stanford University Graduate School of Business Research Paper No. 18-4.

Jacob, M., and R. Michaely, 2017. Taxation and dividend policy: The muting effect of agency issues and shareholder conflicts, *Review of Financial Studies* 30, 3176–3222.

Jensen, M.C., and W.H. Meckling, 1976. Theory of the firm: Managerial behavior, agency costs and ownership structure. *Journal of Financial Economics* 3, 305–360.

Jensen, M.C., 1986. Agency costs of free cash flow, corporate finance and takeovers. *American Economic Review* 76, 323–329.

Kárpáti, D., Renneboog, L., Verbouw, J. 2024, The Family Firm: A Synthesis, Stylized Facts, and Future Research Directions, *Annals of Corporate Governance* 8, 51-157.

Kerr, W., Nanda, R., and M. Rhodes-Kropf, 2014, Entrepreneurship as Experimentation, *Journal of Economic Perspectives* 28, 25–48.

Khan, T., 2006. Company dividends and ownership structure: Evidence from UK panel data, *Economic Journal* 116, 172-189.

Khanna, T., and Y. Yafeh, 2007. Business groups in emerging markets: Paragons or parasites? *Journal of Economic Literature* 45, 331–372.

Lagaras, S., and M. Tsoutsoura, 2015. Family control and the cost of debt: Evidence from the great recession. Working paper, https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=2618560.

La Porta, R., F. Lopez-De-Silanes, and A. Shleifer, 1999. Corporate ownership around the world, *Journal of Finance* 54, 471–517.

Leland, H. and D. H. Pyle, 1977. Informational asymmetries, financial structure, and financial intermediation. *Journal of Finance* 32, 371–387.

Lerner, J., and U. Malmendier, 2013, With a Little Help from My (Random) Friends: Success and Failure in Post-Business School Entrepreneurship, *Review of Financial Studies* 26, 2411–2452.

Lev, B., 1974. On the association between operating leverage and risk, *Journal of Financial and Quantitative Analysis* 9, 627–642.

Lins, K.V., P. Volpin and H. F. Wagner, 2013, Does Family Control Matter? International Evidence from the 2008–2009 Financial Crisis, *Review of Financial Studies* 26, 2583-2619.

Lowry, M., 2023, The Blurring Lines between Private and Public Ownership, ECGI Working Paper Series in Finance Working Paper N° 844/2022

Manso, G., 2016, Experimentation and the Returns to Entrepreneurship, *Review of Financial Studies* 29, 2319-2340.

Maug, E., 1998. Large shareholders as monitors: Is there a trade-off between liquidity and control? *Journal of Finance* 53, 65–94.

Maury, B., 2006. Family ownership and firm performance Empirical evidence from Western European countries, *Journal of Corporate Finance* 12, 321–341.

Mehrotra, V., R. Morck, J. Shim, and Y. Wiwattanakantang, 2013. Adoptive expectations: Rising sons in Japanese family firms. *Journal of Financial Economics* 108, 840–885.

Miller, D., I. Le Breton-Miller, R.H. Lester, and A.A. Cannella, 2007. Are family firms really superior performers? *Journal of Corporate Finance* 13, 829–858.

Miller, M.H., and Modigliani, F., 1961. Dividend policy, growth, and the valuation of shares, *Journal of Business* 34, 411–433.

Modigliani, F., and M.H. Miller, 1958. The cost of capital, corporation finance, and the theory of investments, *American Economic Review* 48, 261–297.

Moeller, S.B., F.P. Schlingemann, and R.M. Stulz, 2004. Firm size and the gains from acquisitions, *Journal of Financial Economics* 73, 201-228.

Moskowitz, T.J., and A. Vissing-Jørgensen, 2002. The returns to entrepreneurial investment: A private equity premium puzzle? *American Economic Review* 92, 745–778.

O'Boyle, E.H. Jr., J.M. Pollack, and M.W. Rutherford, 2012. Exploring the relation between family involvement and firms' financial performance: A meta-analysis of main and moderator effects. *Journal of Business Venturing* 27, 1–18.

OECD/European Commission (2023), The Missing Entrepreneurs 2023: Policies for Inclusive Entrepreneurship and Self-Employment, OECD Publishing, Paris, <a href="https://doi.org/10.1787/230efc78-en">https://doi.org/10.1787/230efc78-en</a>

Pagano, M., 1993. The flotation of companies on the stock market: A co-ordination failure model, *European Economic Review* 37, 1101-1125.

Palmer, J., 1973. The profit-performance effects of the separation of ownership from control in large U.S. industrial corporations. *Bell Journal of Economics* 4, 293–303.

Perez-Gonzalez, F., 2006. Inherited control and firm performance. *American Economic Review* 96, 1559–1588.

Van Praag, M. and A. Raknerud, 2014, The entrepreneurial earnings puzzle, SSB Discussion Papers no. 789.

Roe, M.J., 1994. Strong managers, weak owners: The political roots of American corporate finance. Princeton, NJ: Princeton University Press.

Schmalz, M. C., D. A. Sraer, and D. Thesmar, 2017. Housing collateral and entrepreneurship. *Journal of Finance* 72, 99-132.

Shanker, M.C., and J.H. Astrachan, 1996. Myths and realities: Family businesses' contribution to the US economy – A framework for assessing family business statistics. *Family Business Review* 9, 107–123.

Shleifer, A., and R.W. Vishny, 1997. A survey of corporate governance, *Journal of Finance* 52, 737–783.

Sraer, D., and D. Thesmar, 2007. Performance and behavior of family firms: Evidence from the French stock market. *Journal of the European Economic Association* 5, 709–751.

Strøm, R. Ø., 2019, The Norwegian Gender Balance Law. A reform that failed?, *Annals of Corporate Governance* 4, 1–86.

Sutton, J., 1997. Gibrat's legacy, *Journal of Economic Literature* 35, 40-59.

Tirole, J., 2001. Corporate governance. *Econometrica* 69, 1–35.

Villalonga, B., and R. Amit, 2006a. How do family ownership, control, and management affect firm value, *Journal of Financial Economics* 80, 385–417.

Villalonga, B., and R. Amit, 2006b. Benefits and costs of control-enhancing mechanisms in U.S. family firms, Working paper, Harvard Business School and Wharton.

Villalonga, B., R. Amit, M.-A. Trujillo, and A. Guzman, 2015. Governance of family firms. *Annual Review of Financial Economics* 7, 635-654.

Villalonga, B., and R. Amit, 2020. Family ownership. *Oxford Review of Economic Policy* 36, 241-257.

## Appendix

## Variable definition list

Sample	Definition
Firm	Limited liability firm registered in Norway. If a
	firm has corporate owners, we look up the
	shareholders of those owners to establish
	the ultimate ownership. In the case of
	business groups, we choose the highest level
	in the group where consolidated accounts
	are reported.
Family firms	Firms where an individual or a group of
	individuals related by blood or marriage
	ultimately control more than 50% of the
	equity.
Nonfamily firms	Firms that are not classified as family firms.
Large firms	Firms with sales above NOK 100 mill. (in 2023
	NOK) and more than 100 employees.
Medium firms	Firms with sales between NOK 10 mill. and
	100 mill. (in 2023 NOK) and between 10 and
	100 employees.
Listed firms	Firms that are listed on the Oslo Stock
	Exchange, or business groups that include at
	least one firm listed on the Oslo Stock
	Exchange.
Foreign firms	Firms where foreign companies or individuals
	ultimately control at least 50% of the equity.

Variable	Definition
Sales	Annual revenues in million kroner, inflation
	adjusted (in 2023 kroner).
Total assets	Book value of total assets in million kroner,
	inflation adjusted (in 2023 kroner).
Employment	Number of employees in the firm or group.
	Note that the reporting system changed in
	<u>2015</u> .
Value added	The sum of the firm's earnings and payroll
	expenses, in million 2023 kroner (inflation
	adjusted).
Sales growth	Real growth rate of sales, (Salest – Salest-
	1)/Salest-1, inflation adjusted, winsorized at
	97.5%.

Asset growth	Real growth rate of total assets, (Assets $_t$ – Assets $_{t-1}$ )/Assets $_{t-1}$ , inflation adjusted,
	winsorized at 97.5%.
Employment growth	Growth of the number of employees,
	$(Employment_t - Employment_{t-1})/Employment_{t-1}$
	1, inflation adjusted, winsorized at 97.5%.
Firm age	The age of the firm in years. In the case of
	business groups, it is the age of the oldest
	firm in the group.
Investment rate	The growth rate of fixed assets, adjusted for
	depreciation and writedowns. (Fixed assets $_{t}$ +
	depreciation <sub>t</sub> + writedowns <sub>t</sub> – fixed assets <sub>t-</sub>
	<sub>1</sub> )/fixed assets <sub>t-1</sub> . Winsorized at 97.5%.
Sales volatility	The coefficient of variation (standard
	deviation divided by the mean) of inflation-
	adjusted sales in years t-2, t-1, and t.
Assets per employee	Assets divided by employment.
Value added per employee	Value added divided by employment.
Sales to assets	Sales divided by total assets, winsorized at
	97.5%.
Cash to total assets	The ratio of cash and cash equivalents to
	total assets, winsorized at 97.5%.
Working capital to total assets	The ratio of working capital to total assets,
5	winsorized at 97.5%.
Current ratio	The ratio of current assets to current
	liabilities, winsorized at 97.5%.
Accounts payable days	The ratio of accounts payable to cost of
	goods sold (COGS), winsorized at 97.5%.
Accounts receivable days	The ratio of accounts receivable to sales,
	winsorized at 97.5%.
Inventory days	The ratio of inventory to cost of goods sold
	(COGS), winsorized at 97.5%.
COGS to sales	The ratio of cost of goods sold (COGS) to
	sales, winsorized at 97.5%.
Operating return on assets	The ratio of after-tax operating earnings to
	total assets, winsorized at 2.5% and 97.5%.
Return on equity	The ratio of net income to total net earnings,
notani on oquity	winsorized at 2.5% and 97.5%.
Return on invested capital	The ratio of operating earnings to assets net
	of cash and current debt, winsorized at 2.5%
	and 97.5%.
Earnings per employee	The ratio of net earnings to the number of
	employees, in million 2023 (inflation-
	adjusted) kroner.
Salaries to sales	The ratio of payroll expenses to sales,
	winsorized at 97.5%.
Salary ner employee	The ratio between payroll expenses and the
Salary per employee	
Leverage	number of employees.
Leverage	The ratio of liabilities to total assets.
Net leverage	The ratio of liabilities less cash to total assets
	less cash.

Short-term to long-term debt	The ratio of short-term financial debt to long-
	term financial debt, winsorized at 97.5%.
Interest coverage ratio	The ratio of EBITDA to interest expenses,
	winsorized at 97.5%.
Interest expense to sales	The ratio of interest expenses to sales,
	winsorized at 97.5%.
Dividend payout ratio	The ratio of dividends to (positive) net
	income, winsorized at 97.5%.
Proportion of dividend payers	The proportion of firms with strictly positive
	dividend payments in a group or year.

## Financial indicators for family and nonfamily firms by industry, 2023

The graphs below present financial indicators (means and medians for 2023) by industry for family firms (in blue) and nonfamily firms (in red).



























































































































